Author's response to reviews

Title: Deficiency in immunity to poliovirus type 3: A lurking danger?

Authors:

- Claudia Reinheimer (claudia.reinheimer@kgu.de)
- Imke Friedrichs (imke.friedrichs@kgu.de)
- Holger F Rabenau (rabenau@em.uni-frankfurt.de)
- Hans W Doerr (h.w.doerr@em.uni-frankfurt.de)

Version: 4 Date: 9 December 2011

Author's response to reviews: see over
Dear Editor,

Dear Reviewer #1 and #2,

Thank you very much for the opportunity to revise our manuscript.

We would like to express a special thank you to Reviewer #2 for his or her extensive and helpful linguistic suggestions for improvement.

We hope our corrections and modifications are satisfying and sufficient.

The modifications are red-marked in our manuscript.

Thank you.

Kind regards on behalf of the co-authors

Claudia Reinheimer